\begin{table}%t4
\caption{GELATO's best match templates.}
\label{table2}
\small%\centerline
 {
\begin{tabular}{c c c c c c c}
\hline \hline
SN & Type & Best match  & $QoF$ & Template phase & Template spectrum & Phase reference\\
   &      & SN template &     & (days)         & reference         &                \\
\hline
2002an & II  & 1996bw & 1.7  &  7  $^{*}$  & ASA, \citet{benetti96} & 1995ad, 12d -- \citet{pastorello03}\\
2002bh & II  & 1995V  & 1.4  &  8          & ASA                    & \citet{fassia98}\\
2002cs & Ia  & 2004dt & 1.6  & --7          & \citet{altavilla07}    & \\
2002dg & Ib  & 1998dt & 1.7  &  15 $^{**}$ & ASA                    & \citet{matheson01}\\
2002dm & Ia  & 1994ae & 9.1  &  92         & ASA                    & \\
2002ej & II  & 1995ad & 2.9  &  24         & \citet{pastorello03}   & \\
2002hg & II  & 1999em & 2.4  &  41         & \citet{elmhamdi03}     & \\
2002hm & II  & 1998ce & 3.8  &  10 $^{*}$  & ASA, \citet{patat98}   & 1995ad, 12d -- \citet{pastorello03}\\
2002hy & IIb & 2001gh & 1.0  &  11 $^{*}$  & ASA, \citet{altavilla01}         & 1993J, 4d -- \citet{barbon95}\\
2003hg & II  & 1995V  & 2.1  &  8          & ASA                    & \citet{fassia98}\\
2003hn & II  & 1995ad & 2.4  &  10         & \citet{pastorello03}   & \\
2003ie & II  & 1998A  & 1.2  &  37         & \citet{pastorello05a}  & \\
2004G  & II  & 1993S  & 2.4  &  90 $^{*}$  & ASA                    & 1995ad, 100d -- \citet{pastorello03}\\
2004aq & II  & 1991al & 2.9  &  25 $^{*}$  & ASA                    & 2001du, 18d -- \citet{smartt03}\\
2004bs & Ib  & 1998dt & 2.0  &  17 $^{**}$ & ASA                    & \citet{matheson01}\\
2004cc & Ic  & 1994I  & 1.4  &  10         & \citet{filippenko95}   & \\
2004dg & II  & 2001du & 4.8  &  18         & \citet{smartt03}       & \\
2004dk & Ic  & 2004aw & 1.4  &  4          & \citet{taubenberger06} & \\
2004dn & Ic  & 2004aw & 1.4  &  4          & \citet{taubenberger06} & \\
2004fe & Ic  & 1994I  & 1.9  & --3          & \citet{filippenko95}   & \\
2004go & Ia  & 1996X  & 2.2  &  24         & \citet{salvo01}        & \\
2005G  & Ia  & 1994D  & 3.5  &  11         & \citet{patat96}        & \\
2005H  & II  & 2002gd & 1.0  &  6          & \citet{pastorello03}   & \\
2005I  & II  & 2003gd & 3.3  &  101        & ASA                    & \citet{hendry05}\\
2005N  & Ib  & 1990I  & 2.1  &  88         & \citet{elmhamdi04}     & \\
2005V  & Ic  & 2004aw & 1.3  &  22         & \citet{taubenberger06} & \\
2005ab & II  & 1997du & 1.5  &  26 $^{*}$  & ASA, \citet{patat97}   & \\
2005ai & Ia  & 1994D  & 2.7  &  24         & \citet{patat96}        & \\
2005br & Ib  & 1997X  & 1.9  &  40         & ASA                    & 1990U, 48d -- \citet{piemonte96}\\
2005bs & Ia  & 1996X  & 3.3  &  31         & \citet{salvo01}        & \\
2005cb & Ic  & 1994I  & 2.5  &  1          & \citet{filippenko95}   & \\
2005ce & Ib/c& 1996aq & 2.0  &  5 $^{*}$   & ASA                    & 1994I, 10d -- \citet{filippenko95}\\
2005de & Ia  & 2005cf & 2.8  & --5          & \citet{garavini07}     & \\
2005dv & Ia  & 2002bo & 2.9  &  0          & \citet{benetti04}      & \\
2005dz & II  & 2007T  & 1.4  &  4 $^{*}$   & ASA, \citet{benetti07} & 2002gd, 6d -- \citet{pastorello03}\\
2005kl & Ic  & 2004aw & 1.2  &  4          & \citet{taubenberger06} & \\
\hline
\end{tabular}}
\medskip
Notes: phases are relative to the $B$-maximum epoch for type-Ia and Ib/c SNe and to explosion epoch for type-II SNe, $^{*}$~spectral epoch relative to the discovery date. In these cases no reference of the explosion epoch is found and, when available, the second best fitting template spectrum with
phase relative to the explosion epoch is reported in the ``Phase reference'' column, $^{**}$~relative to the $R$-maximum.
\end{table}